Author's response to reviews

Title: Distribution of the FY*BES and RHCE*ce(733C>G) alleles in an Argentinean population: implications for transfusion medicine

Authors:

Carlos M Cotorruelo (ccotorru@fbioyf.unr.edu.ar)
Silvana V Fiori (silvanafiori@hotmail.com)
Silvia E Garcia Borras (sigarcia@fbioyf.unr.edu.ar)
Liliana L Racca (lracca@fbioyf.unr.edu.ar)
Claudia S Biondi (cbiondi@fbioyf.unr.edu.ar)
Amelia L Racca (aracca@fbioyf.unr.edu.ar)

Version: 4 Date: 1 May 2008

Author's response to reviews:

Dear Sirs,

after checking over the manuscript I made the following minor changes:

1) e-mail address of the second author (SVF: silvanafiori@hotmail.com is the correct address).
2) the word assign (page 23, line 5) was replaced by assigned.

Thank you very much,

Kindest regards,

Dr. Carlos Cotorruelo